Author's response to reviews

Title: Screening for low bone mass with quantitative ultrasonography in a community without dual-energy X-ray absorptiometry: population-based survey

Authors:

Nan-Ping Yang (g880703@ym.edu.tw)
Ian Jen (ianjen@ym.edu.tw)
Shao-Yuan Chuang (d48907004@ym.edu.tw)
Shui-Hu Chen (g39107003@ym.edu.tw)
Pesus Chou (pschou@ym.edu.tw)

Version: 8 Date: 25 February 2006

Author's response to reviews:

Dear Editor: 1. The references have been recchecked and corrected. 2. The list of authors are exactly. The e-mail address of all the authors has been added in the manuscript. 3. The abstract in the manuscript has deleted those key-words.